Wilfrid Laurier University

Scholars Commons @ Laurier

Biology Faculty Publications

Biology

2009

A Cost of Sexual Attractiveness to High-Fitness Females

Tristan A.F. Long Wilfrid Laurier University, tlong@wlu.ca

Alison Pischedda University of California -- Santa Barbara

Andrew D. Stewart University of California -- Santa Barbara

William R. Rice University of California -- Santa Barbara

Follow this and additional works at: https://scholars.wlu.ca/biol_faculty

Recommended Citation

Long, Tristan A.F.; Pischedda, Alison; Stewart, Andrew D.; and Rice, William R., "A Cost of Sexual Attractiveness to High-Fitness Females" (2009). *Biology Faculty Publications*. 24. https://scholars.wlu.ca/biol_faculty/24

This Article is brought to you for free and open access by the Biology at Scholars Commons @ Laurier. It has been accepted for inclusion in Biology Faculty Publications by an authorized administrator of Scholars Commons @ Laurier. For more information, please contact scholarscommons@wlu.ca.

A Cost of Sexual Attractiveness to High-Fitness Females

Tristan A. F. Long[#], Alison Pischedda, Andrew D. Stewart, William R. Rice

Department of Ecology, Evolution, and Marine Biology, University of California Santa Barbara, Santa Barbara, California, United States of America

Abstract

Adaptive mate choice by females is an important component of sexual selection in many species. The evolutionary consequences of male mate preferences, however, have received relatively little study, especially in the context of sexual conflict, where males often harm their mates. Here, we describe a new and counterintuitive cost of sexual selection in species with both male mate preference and sexual conflict via antagonistic male persistence: male mate choice for high-fecundity females leads to a diminished rate of adaptive evolution by reducing the advantage to females of expressing beneficial genetic variation. We then use a *Drosophila melanogaster* model system to experimentally test the key prediction of this theoretical cost: that antagonistic male persistence is directed toward, and harms, intrinsically higher-fitness females more than it does intrinsically lower-fitness females. This asymmetry in male persistence causes the tails of the population's fitness distribution to regress towards the mean, thereby reducing the efficacy of natural selection. We conclude that adaptive male mate choice can lead to an important, yet unappreciated, cost of sex and sexual selection.

Citation: Long TAF, Pischedda A, Stewart AD, Rice WR (2009) A Cost of Sexual Attractiveness to High-Fitness Females. PLoS Biol 7(12): e1000254. doi:10.1371/journal.pbio.1000254

Academic Editor: Russell Bonduriansky, University of New South Wales, Australia

Received May 11, 2009; Accepted October 30, 2009; Published December 8, 2009

Copyright: © 2009 Long et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by grants to WRR from the National Science Foundation (DEB-0128780 and DEB-0410112). TAFL held a Natural Science and Engineering Research Council of Canada (NSERC) Postdoctoral Fellowship and received support in the form of a University of California Santa Barbara Harvey L. Karp Discovery Award. AP acknowledges NSERC for a Postgraduate Scholarship. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: tristan.long@utoronto.ca

¤ Current address: Department of Ecology and Evolutionary Biology, University of Toronto, Toronto, Ontario, Canada

Introduction

Historically, most studies of mate choice have focused on mate preference by females, because this sex typically has higher levels of parental investment and lower variance in realized fitness [1-4]. Mate choice by males, however, is a common feature of many species [5-8], yet its adaptive consequences are far less commonly considered [7,9], and are typically only studied in species with reversed sex roles [10-12]. Here, we focus on species with "typical" sex roles that also experience sexual conflict due to antagonistic male persistence (e.g., unrelenting courtship and repeated mating attempts) that arises because the optimal outcomes of mating interactions often differ for males and females [13,14]. It is well established that females can suffer substantial fitness costs from receiving too much male attention [15]. For instance, male sexual persistence is an important source of female mortality and/or reduced fecundity in a number of species, e.g., frogs, Crinea georgiana [16]; toads, Bufo bufo [17]; feral sheep, Ovis aries [18]; lizards, Lacerta vivipara [19]; ducks, Anas platyrhynchos [20]; orangutans, Pongo pygmaeus [21]; water striders, Gerris odontogaster [22]; and fruit flies, Drosophila melanogaster [23,24]. In this study, we focus on a different harmful consequence of females being subject to male persistence that only occurs when males evolve a mate preference for high-fecundity females: a reduced rate of adaptation.

Model

We first develop a graphical model in which a single quantitative trait is a reliable, direct indicator (rather than an indirect indicator, like a costly ornament) of a female's "intrinsic" fecundity (i.e., fecundity in the absence of costly male persistence). For example, in a wide diversity of taxa, variation in female fecundity is strongly correlated with body size [2,4] because larger females have more resources to invest in fecundity. Henceforth, we arbitrarily assume that a female's body size is the phenotypic trait correlated with fecundity, but our logic applies to other indicator traits that directly influence her fecundity, such as parasite load [25] or abdomen size in many insects [7]. Males are expected to evolve a mating preference for larger females whenever this preference increases their own lifetime reproductive success [14]. Such an adaptive male mate preference will cause larger, intrinsically high-fecundity females, to receive more antagonistic male persistence, compared to smaller, intrinsically low-fecundity females. The fitness consequences of this relationship will depend upon how female resistance to male-induced harm scales with the indicator trait (in this case, body size). Assuming that a female's resistance to the harmful male persistence does not rise sufficiently fast with increasing body size, the male preference should reduce the fecundity of large females and increase that of small females, thereby reducing the standing variance in fitness (Figure 1). As a result, the selective advantage of any beneficial genetic variation that makes females more competitive for limiting resources, and hence more fecund, will experience a smaller selective advantage than if harmful male persistence was randomly applied to females throughout the population. Such nonrandom male persistence will cause adaptive evolution in females to be slowed whenever female fecundity is: (i) heritable, (ii) genetically correlated with the indicator trait, and (iii) a major determinant of her lifetime fitness that does not strongly trade off with her other fitness components. A reduced rate of adaptive evolution by females can also be

Author Summary

In many species, females are frequently subject to harassing courtship from males attempting to mate with them. These persistent male behaviors can result in females incurring substantial direct fitness costs. We set out to examine how these costs may influence adaptive potential in a species that also exhibits male mate choice, i.e., a preference by males for females exhibiting certain traits. We found that harmful courtship behaviors were directed predominantly towards females of greater reproductive potential (and away from females of lesser potential), resulting in a reduction in the variation of lifetime reproductive successes among females in the population. This change in distribution of realized fitnesses represents a previously unappreciated consequence of sexual conflict-adaptive male mate preference can slow the rate of accumulation of beneficial mutations and speed the rate of accumulation of harmful mutations, thereby creating a "sexual conflict adaptive load" within a species.

deduced from Fisher's fundamental theorem [26], so long as the male-induced reduction in the phenotypic variation in female fecundity also leads to a reduction in the additive genetic variation among females. Furthermore, when there is a positive genetic correlation for fitness between females and males, adaptive male mate choice is expected to reduce the rate of adaptation in both sexes. Although a counteracting effect could occur if male preference for high-fecundity females increases the variance in male fitness, or if male preferences lead to positive assortative mating for fitness, here we focus on female fitness and the potential for male mate preferences to reduce its heritable variation.

The conclusion that adaptive male choice leads to a reduced rate of adaptation by females can also be deduced by focusing on mutations at a single arbitrary locus. Let the mutation rate to new beneficial mutations be U_{Ben} and the selective advantage of the mutation, expressed as a selection coefficient and averaged

Figure 1. Effect of mate choice on the distribution of female fitness in species with antagonistic persistence. When males direct their antagonistic persistence more towards higher-fitness females, the tails of the female fitness distribution are predicted to regress towards the mean (dotted line and shaded distribution) compared to the case where male persistence is randomly applied to all females in the population (solid line and open distribution). doi:10.1371/journal.pbio.1000254.g001

across the sexes, be *s*. Assuming approximate additivity (i.e., little dominance), the probability of the mutation becoming fixed can be calculated using the diffusion approximation [27] as $2s(N_e/N)$, where N is the population size and N_e is the effective population size. With recurrent mutation to new beneficial mutations, the rate of advance of adaptive evolution is approximated by:

Rate of evolution =
$$U_{\text{Ben}} \times prob_{\text{Fixation}}$$
 (1)

$$\approx U_{\text{Ben}} \times 2s(N_e/N)$$
 (2)

If we partition selection between the sexes and let the selective advantage of a mutation be s_{σ} in males and s_{φ} in females then,

Rate of evolution =
$$U_{\text{Ben}} \times 2(s_{\text{Q}} + s_{\text{C}})(N_e/N)/2$$
 (3)

$$\approx U_{\rm Ben} \times \left(S_{\rm Q} + S_{\rm O} \right) (N_e/N) \tag{4}$$

Next, we assume that the expression of the beneficial mutation also increases the attractiveness of females to males (e.g., as a result of increasing her body size), so that those females expressing the beneficial mutation receive an excess of antagonistic male persistence. We express this cost with an additional selection coefficient $s_{\text{Qbiased-persist}}$, which is applied only to females,

Rate of evolution
$$\approx U_{\text{Ben}} \times \left(\left[S_{\text{Q}} + S_{\text{Qbiased-persist}} \right] + S_{\text{C}} \right) (N_e/N)$$
 (5)

Comparison of Equations 4 and 5 demonstrates that the rate of adaptive evolution will always be slower whenever males bias their



antagonistic persistence towards fitter females and cause $s_{\text{Qbiased-persist}}$ to be negative; i.e., when there is adaptive male mate choice and increased male persistence is harmful to females. Increased male persistence directed towards more fecund females that express the beneficial allele reduces the selective advantage of those females and thereby reduces the variance in fitness among females in the population.

Predictions and Assumptions

The primary prediction from our models is that, in species with antagonistic male persistence, adaptive male mate preference leads to a "cost of being an attractive female." This cost reduces the selective advantage of females expressing more beneficial genetic variation (and hence are larger, on average) and increases the fitness of females expressing less of this variation (and hence are smaller, on average). Put more simply, adaptive male mate preference causes the tails of the population's distribution of female lifetime fecundity to regress towards the mean (Figure 1). This prediction is contingent on four assumptions that must be met in order for our model to operate: (i) lifetime fecundity and net fitness are strongly genetically correlated, (ii) body size and fecundity are positively correlated, both phenotypically and genetically, (iii) more antagonistic male persistence is directed towards females with higher intrinsic fecundity (i.e., potential fecundity in the absence of costly male persistence), and (iv) female resistance to maleinduced harm does not rise sufficiently fast with increasing body size. We tested the major prediction of the model, and its underlying assumptions, using a laboratory population of the model species D. melanogaster. In this population, assumption (i) is well established [28,29], so here we focus on testing whether our population meets assumptions ii-iv, before experimentally



assessing whether male mate preference for high-fitness females causes the tails of the distribution of lifetime fecundity to regress towards the mean.

Results

Assumption (*ii*)—A Positive Genetic and Phenotypic Correlation between Body Size and Lifetime Fecundity

Joint measures of female body size and lifetime fecundity in our base population (LH_M) of D. melanogaster indicated that these two phenotypic traits are strongly correlated. As predicted from past studies of many taxa [2,4,30], fecundity was higher in large females compared to small females (Figure 2). This result was found both when females experienced minimal exposure to males (mean \pm standard error [SE]: large females, 27.3 \pm 1.59; small females, 18.42 ± 1.07 ; *t*-test t = 4.64, df = 98, p < 0.0001; *p*values reported throughout the manuscript are two-tailed) and when male exposure was continuous (large females, 16.5 ± 1.06 ; small females, 12.62 ± 0.72 ; *t*-test t = 3.01, df = 98, p < 0.003). We tested for a genetic correlation between body size and fecundity in a separate study in which two populations each were artificially selected for either large or small body size. After 83 generations of divergent selection, lifetime fecundity was significantly higher in the lines selected for large body size compared to the lines selected for small body size (mean \pm SE: large females, 31.38 ± 1.81 ; small females, 15.25 ± 1.58 ; *t*-test t = 6.69, df = 2, p = 0.02). Since body size was the only target of artificial selection, the large divergence in fecundity between treatments demonstrates a strong positive genetic correlation between body size and fecundity, a result consistent with other research [31].

Figure 2. Effect of the extent of male exposure on female fecundity. Mean (±SE) lifetime fecundity (number of eggs produced) when individual large (shaded bars) and small (open bars) females experience minimal or continuous exposure to males. doi:10.1371/journal.pbio.1000254.g002

Assumption (*iii*)—Antagonistic Male Persistence Is Directed More towards Females with Higher Intrinsic Fecundity

To test this assumption, we first measured how the persistence (courtship behaviour) of individual males was allocated between two nonvirgin females (differing in eye colour phenotype, brown or red, for ease of individual identification). We performed a two-way ANOVA on the amount of persistence behaviour directed towards each female, with the body size of that "target" female (large or small), the body size of the competitor female present in the test tube (large or small), the eye colour of the target female (red or brown), and all possible interactions as predictor variables. This analysis was significant overall ($F_{7,232} = 7.50$, p < 0.0001), with significant effects of the target female body size ($F_{1,232} = 38.37$, p < 0.0001) and the body size of the competitor female $(F_{1,232} = 13.53, p = 0.0003)$, but no effects of eye colour $(F_{1,232} = 0.41, p = 0.52)$, or any of the interactions (all p > 0.60). When individual males were housed with two nonvirgin females differing in body size, males directed more persistence towards the larger female than towards the smaller female (paired t-tests, $p \leq 0.0002$, Figure 3). When males were housed with two nonvirgin females of similar body size (both small or both large), the levels of male persistence directed towards the red- and brown-eyed females were not significantly different (paired *t*-tests, $p \ge 0.50$, Figure 3). Further evidence of a male mate preference for nonvirgin females of larger body size was obtained from mating assays conducted under conditions that more closely mimicked the normal culture environment of the LH_M population (16 males combined with 16 females during the "adult competition" phase of the life cycle [28,32]). When presented with a choice of nonvirgin females differing in body size, males mated with large-bodied females at a greater rate than with small-bodied females (generalized linear model [GLM] with binomial error terms; $p_{\text{consensus}} = 1.17 \times 10^{-5}$, Replicate 1: $\chi^2_{1,18} = 5.90$, p = 0.015; Replicate 2: $\chi^2_{1,74} = 22.87$, p < 0.0001; Figure 3). These remating results are unlikely to have arisen from large females possessing a greater receptiveness to male courtship effort because males kept under "no-choice" mating conditions (where either only large or only small nonvirgin females were present) mated with small females more frequently than with large females (GLM with binomial error terms; $p_{\text{consensus}} < 1 \times 10^{-6}$, Replicate 1: $\chi^2_{1,28} = 14.79$, p = 0.0001; Replicate 2: $\chi^2_{1,28} = 7.16$, p = 0.0075; Figure 4).

Assumption (*iv*)—Female Resistance to Male-Induced Harm Does Not Rise Sufficiently Fast with Increasing Body Size

To test this assumption, we compared the reduction in lifetime fecundity of large and small females when they were either minimally or continuously exposed to males (Figure 2). Continuous male exposure harmed large females more than small females (two-way ANOVA, interaction between body size and male exposure, $F_{1,196} = 4.75$, p = 0.031), indicating that larger females were not more resistant to the harmful male persistence that they received.

Prediction—Male Mate Preference for High-Fitness Females Causes the Tails of the Distribution of Lifetime Fecundity to Regress towards the Mean

Finally, we tested the model's key prediction by comparing the mean fecundities of large- and small-bodied females used in the choice and no-choice mating assays described earlier. In the nochoice assays, where all females were either of large or small body size, male preference for large female body size could not cause



Figure 3. Male mate persistence towards females of similar or differing body size. Mean (±SE) number of male persistence behaviours directed towards either brown-eyed (open bars) or red-eyed (shaded bars) females of large or small body size. doi:10.1371/journal.pbio.1000254.g003



Figure 4. Remating rates under different male-choice environments. Mean (\pm SE) proportion of eight large (closed circles) or eight small (open circles) nonvirgin females that mated over a 24-h period when housed in vials with 16 males and eight similar-sized females (no-choice) or 16 males and eight randomly selected females (choice). Data are from two replicate assays. Number of replicated vials per assay is shown in brackets. doi:10.1371/journal.pbio.1000254.g004

them to direct their antagonistic persistence away from smaller females and towards larger females. In contrast, in the choice assays, where females of different body sizes were simultaneously present, a redirection of antagonistic male persistence towards larger females was possible. We found that the difference in the mean fecundities of large- and small-bodied females was smaller when males could direct their antagonistic persistence towards large females (Figure 5, $p_{\rm consensus} = 0.012$, interaction tests for each replicate: F_{1,46} = 2.79, p = 0.1 for the smaller, first replicate; F_{1,101} = 4.92, p = 0.03 for the larger, second replicate).

Discussion

The results of our male mate preference tests clearly demonstrate that males have mate preferences for larger nonvirgin females-a result consistent with earlier work on virgin females [33]. Rather than displaying an "undiscriminating eagerness" [34] to mate, when given a choice between females differing in body size, male D. melanogaster preferred to court and mate with large, high-fecundity females over small, low-fecundity females. Given the significant fecundity differences associated with female body size described above, this mate preference is likely to be adaptive from the male's perspective, as mating with larger, more fecund females is likely to yield greater direct, as well as indirect [35], benefits. It is unlikely that this male mate preference is adaptive from the female's perspective, as several studies have established that chronic male persistence in the LH_M population is very harmful to females, and it is not sufficiently compensated by indirect genetic benefits [36-38]. Our experiments demonstrate that larger females receive more harmful male persistence but do not reveal the specific mechanism by which

this harm accrues. Further work will be needed to resolve the degree to which this increased harm is due to harassment during courtship [24], damage associated with copulation [39], and/or the activity of products transferred in the male's seminal fluid [23,40].

Having experimentally ascertained that the LH_M population of D. melanogaster satisfied all the assumptions necessary in which to test the key prediction our model, we were able to meaningfully assess the fitness consequences of adaptive male mate preferences. When males had the ability to bias their antagonistic persistence towards large-bodied females, we saw a decrease in the mean fecundity of these preferred females, compared to those large females that were in an experimental environment where all females were of similar size, and biases of antagonistic male persistence were not possible. In contrast, small-bodied females were, on average, able to realize relatively higher fecundities when they were housed with larger females (which, our study indicates, were attracting more harmful male persistence) than they were when they were housed in an environment in which males had no other choice of mates. Although our study found that males directed more courtship towards large females and also mated them more frequently, both of which can be harmful in and of themselves [24], the observed cost to large females might also have occurred because large females were mated, on average, to more harmful males [30,41] than were smaller females. Irrespective of the mechanism of this cost, together these assays revealed how male mate preferences will ultimately cause the tails of the distribution of fecundity to regress towards the mean. Since adult lifetime fecundity is strongly correlated with lifetime fitness in females of the LH_M population [42], this male-driven sexual selection is expected to reduce the rate of adaptive evolution of any



Figure 5. Effects of male mate persistence and choice on female fecundity. Interaction plots comparing mean (\pm SE) fecundities (number of eggs produced) of large (closed circles) and small (open circles) females from the two replicates of the mate choice and no-choice assays (described in legend of Figure 4).

doi:10.1371/journal.pbio.1000254.g005

trait that is positively correlated with female body size. It is common for deleterious mutations to reduce body size in D. melanogaster [43], and it is reasonable to assume that many beneficial mutations will cause their carriers to be more competitive as larvae, allowing them to garner more resources during the larval competition phase of their life cycle and become larger, more fecund, adults. As a consequence, male mate preference for larger females is expected to commonly interfere with both progressive evolution and to increase the population's mutational load by interfering with purifying selection. For example, suppose that environmental change led to selection for alleles conferring higher desiccation tolerance. If more desiccationtolerant females had a competitive advantage such that they grew to a larger size prior to reproduction (e.g., [44]), then a male preference for these females would reduce their relative fecundity and increase that of smaller, less desiccation-tolerant females. As a result, the population may be less responsive to environmental change, become an inferior competitor species, and be at a greater risk of extinction.

Collectively, our results support our model's key prediction that male mate preference for high-fitness females reduces the selective advantage of larger, more fecund females and increases that of smaller, less fecund females. This finding, obtained in a laboratory population, is likely to apply to natural populations for two reasons. First, the study was done on a large, outbred population that has been maintained in a competitive laboratory environment, at continuous large size, for over 400 generations [28,32]. Over this period of time, the opportunity for adaptation to the laboratory environment should have been substantial, permitting the flies to be experimentally assayed under conditions to which they are highly adapted. Second, we measured natural variation in body size, rather than inducing extreme body size variation via nutritional deprivation and/or excessive larval crowding. This was accomplished using a sieve shaker device (developed by ADS and WRR), which enabled us to quickly sort thousands of adult flies based on natural variation in their body size, and obtain the largest and smallest individuals to use in our experiments. Flies from these two body size groups differed markedly in fecundity, with the larger females producing over 30% more eggs than small females under both minimal and continuous male exposure conditions. Although our assays of male mate preference support a directional preference for large-bodied females, in one assay (Figure 2), males could only choose between females of large and small body size. Thus, there is the possibility that the true male preference function favours females of intermediate size. However, in our second assay (Figure 3), males were able to choose between large or small females versus random females (average), and these data support the conclusion that male preference is monotonic for larger females.

Our model of adaptive male mate choice in the context of harmful male persistence has important limitations. First, we have implicitly assumed that the increased male persistence (directed toward larger, more intrinsically fecund females) does not cause larger females to have lower than average fecundity. Second, male condition may be more variable in nature compared to the laboratory, and condition-specific patterns of male persistence could either enhance or reduce the bias of male persistence toward larger females. Third, we have ignored complicating factors such as size-assortative mating interactions, e.g., smaller females receiving persistence predominantly from smaller or poorcondition males. Fourth, we have assumed that male mate choice is based on a female trait that directly influences her fecundity, such as body size. Theory predicts that this type of male mate preference will lead to a monotonic preference for larger females [45,46]. However, when the preferred female trait is a costly indicator of fecundity, such as an energetically expensive ornament, then males can evolve to prefer intermediate trait values in females [45,46], and our model would not apply. Fifth, our model may not apply to species where females obtain direct net benefits from increased mating rates, such as those with nuptial feeding [47]. Lastly, we have assumed a static male preference and female indicator trait. In many contexts, these two traits can be expected to coevolve, and this dynamic is not included in our model. Nonetheless, our empirical work suggests that the requisite conditions for the model to operate, at least transiently, can feasibly be achieved.

Our finding of harmful effects of adaptive male mate choice represents a previously unappreciated cost of sexual reproduction in species with antagonistic male persistence. Rather than simply showing that male-induced harm reduces overall female fecundity, we have shown that biases in the distribution of this harm among mates reduces the selection differential between females with intrinsically high and low fecundity. This reduced efficacy of natural selection will retard a population's rate of adaptive evolution and increase both its equilibrium mutational load and its stochastic accumulation of harmful mutations. The cost of adaptive male mate choice, however, only applies when males can reliably ascertain a female's fecundity using a trait that is heritable and correlated with heritable fitness variation. In Drosophila, female body size represents such a trait since it is influenced by both genotype [48,49] and a number of environmental factors (including temperature, nutrition and larval crowding conditions [50]), and responds rapidly to directional selection. In species with little or no heritability for body size, however, an adaptive cost of male preference for high-fecundity females would not apply. Nonetheless, given the prevalence of male mate preferences [7], this new cost that we describe may be a widespread evolutionary phenomenon. For this reason, it should be considered in the broader context of the ongoing debates over the interfering or reinforcing role that sexual selection plays in the process of adaptation, and whether sexual selection increases or decreases the risk of extinction of populations and species [51].

Materials and Methods

Experimental Animals and Female Size-Sorting Technique

For all male-female interaction assays, we used D. melanogaster adults obtained from the wild-type LH_M population [28,29] or from a replicate population $(LH_M - bw^D)$ in which a dominant brown-eyed marker (bw^D) had been introgressed through repeated backcrossing into the LH_M genetic background. The LH_M population is maintained on a 14-d culture cycle with a 12-h L:12-h D diurnal cycle at 25°C in humidity-controlled incubators. Briefly, each generation begins with eggs placed in 56 "juvenile competition" vials (150–200 eggs per vial; each vial containing 10 ml of cornmeal/molasses medium). After 11.25 d, emerging adults are lightly anesthetized with CO₂, mixed among vials, and transferred to "adult competition" vials (16 pairs of males and females per vial), which are seeded with 6.4 mg (dry weight) of live yeast. After 2 d of adult competition, the flies are transferred to "oviposition" vials, and then discarded after laying eggs for 18 h. The eggs laid in these oviposition vials are culled to a density of 150-200 eggs per vial and become the "juvenile competition" vials of the next generation. Because only eggs from the oviposition phase of the life cycle are used to propagate the next generation, and populations have been consistently maintained under these culture conditions for over 400 generations, the number of eggs laid during the 18-h oviposition phase represents a meaningful measure of lifetime fecundity in these populations. As such, experiments were designed to mimic these culture conditions as closely as possible. Detailed culturing protocols for these large populations (adults n>1,800 per generation for LH_M and n>1,300per generation for LH_M- bw^D) can be found elsewhere [28,29]).

We altered the quality of potential female mates by collecting females of differing adult body size, a phenotypic trait that is frequently positively correlated with fecundity [2,30,52]. We collected flies from the ends of the normal distribution of body sizes that are produced under typical lab culture conditions. Flies were sorted by size with the use of a sieve shaker device (Gilson Performer III, Gilson Company) which mechanically separates anesthetized flies on the basis of their ability to pass through a series of 20 electroformed sieves, in which the diameter of the holes in each sieve was 5% larger than the diameter of the holes of the sieve below (diameter of top sieve holes = $1,685 \mu m$; diameter of bottom sieve holes = $800 \mu m$). Flies were placed into the column (under light CO₂ anaesthesia), and were agitated at a rate of 3,600 vibrations min⁻¹ for 2 min. By using this technique, it was possible to quickly sort hundreds of flies simultaneously on the basis of their body size. For all experiments, "small" flies were defined as those that were small enough to pass through the 1,095-µm diameter sieve, whereas "large" flies were those that were too large to pass through the 1,281-µm diameter sieve.

Assessment of the Phenotypic Correlation between Body Size and Fecundity in Females, and Quantification of the Relationship between Body Size and Female Resistance to Male-Induced Harm

To assess the phenotypic correlation between body size and fecundity, we collected adult flies from the LH_M population as they eclosed as virgins on day 9 of their life cycle. Flies were separated by sex, and on the following day, females were sorted by size using the sieve sorter protocol described above. One hundred female flies each of large and small body size were then placed individually (under light anaesthesia) into small test tubes that had been seeded with 0.4 mg of yeast (the amount of yeast per female experienced under normal culture conditions). Into each of these vials, three adult males were placed for a period of 2 h, during which time all virgin females were observed to have mated once. Males were then removed randomly from half of the vials to create 50 adult competition vials with minimal male exposure and 50 with continuous male exposure for each female body size category. Maintaining flies under these two conditions allows us to confirm that there is an intrinsic difference in fecundity between females of different sizes that is independent of the negative net fitness effects of continuous male presence. Matching the normal culturing protocol of the flies, vials were returned to the incubator for an additional 2 d, at which time flies were transferred to oviposition vials containing fresh medium (with a scored surface to encourage oviposition) for a period of 18 h before being discarded. The number of eggs laid in each vial was counted, and mean fecundities were compared using t-tests for females differing in size in each male-exposure treatment. The complete dataset was also used to test the assumption that female resistance to male-induced harm does not rise sufficiently fast with increasing body size, by examining whether or not female flies of one size were harmed more by continuous male exposure.

Assessment of the Genetic Correlation between Body Size and Fecundity in Females

In order to verify that there was a genetic correlation between body size and fecundity, we assessed the fecundity of females obtained from populations of D. melanogaster that are part of an ongoing experimental evolution project (of ADS and WRR) in which females had been artificially selected for either large or small body size using a size-sorting procedure similar to that described above. These populations are otherwise cultured in a manner similar to the LH_M population from which they were all originally derived. At the time of the assay, the artificial selection had been operating for 83 generations in each of two replicate populations per treatment, and there had been considerable divergence in body size (mean female diameter $[\mu m] \pm SE$: large treatment, $1,218.5 \pm 37.67$; small treatment, 786.7 ± 43.5 ; *t*-test t = 7.51, df = 2, p < 0.01). For this assay, 72 virgin females were obtained at random from each of the four experimental populations. On day 11 of their life cycle, these females were placed in adult competition vials in groups of 16, along with 16 males taken randomly from the LH_M population, for a period of 2 h, during which time all females were observed to have mated once. Males were removed from the vials, and after 2 d in the incubator, females were transferred to individual oviposition vials containing fresh medium (with a scored surface) for a period of 18 h before being discarded. The number of eggs laid in each vial was counted and the mean fecundity of the two replicates of each treatment was compared using a *t*-test (with population as the unit of replication). Since the selected trait in these experimental populations was body size, any consistent change in fecundity between the two treatments must be due to a genetic correlation between the two traits.

Behavioural Assay of Male Mate Persistence towards Females Differing in Body Size

In order to test whether males have mate preferences, a series of behavioural assays were conducted. Nonvirgin flies from both the LH_M and LH_M - bw^D populations were collected on day 11 of their life cycle, and females were sorted by size to isolate large- and small-bodied individuals. Pairs of female flies differing in eye colour (to aid individual identification) were placed into small, adult competition vials (test tubes) in all possible combinations of body size. After a 1-h anaesthesia-recovery period, a single unanaesthetized adult male fly was added to each test tube, which were then placed on their sides in a well-lit room. Over the course of 11 sessions, spaced 40 min apart, the male in each test tube was observed. Male persistence behaviour was defined as being located within 5 mm of a female and oriented towards her [53-55]. Data on the frequency of the male persistence behaviour was collected for each type of female in each treatment. A total of 30 replicate test tubes per treatment were scored.

Assays of Remating Rates for Females Differing in Body Size under "Choice" and "No-Choice" Environments

In these assays, nonvirgin adult female LH_M flies were collected on day 11 of their life cycle and sorted by size (see above) to isolate

References

- Bateson P, ed (1983) Mate choice. Cambridge (United Kingdom): Cambridge University Press. 462 p.
- Andersson MB (1994) Sexual selection. Princeton (New Jersey): Princeton University Press. 599 p.
- Jennions MD, Petrie M (2000) Why do females mate multiply? A review of the genetic benefits. Biol Rev Camb Philos Soc 75: 21–64.
- Clutton-Brock TH, ed (1988) Reproductive success: studies of individual variation in contrasting breeding systems. Chicago (Illinois): University of Chicago Press. 538 p.

large and small individuals. Females were then placed into one of two types of adult competition vials (a vial containing fresh medium seeded with 6.4 mg of live yeast). In the first, choice experiment, either eight large or eight small red-eyed LH_M females were placed into an adult competition vial along with eight randomly collected LH_M - bw^D females and 16 LH_M - bw^D males. In the second, no-choice treatment, either 16 large or 16 small redeved LH_M adult females were placed into an adult competition vial along with 16 LH_M -bw^D males. These vials were kept in the incubator (on their sides) for 24 h, at which time males were removed. The vials, containing females only, were then returned to the incubator for an additional 24 h. Remating rates were assaved by placing all females into individual oviposition vials (test tubes) containing fresh, scored medium for the purpose of measuring the paternity of her offspring. Eighteen hours later, the adult flies were discarded, and the test tubes containing eggs were incubated for 11 d. At this time, the presence and number of red-eyed and brown-eyed progeny in each brood were scored to ascertain whether the female had remated. The proportion of females in each adult competition vial that produced brown-eyed offspring (indicating a remating event) was recorded. To examine remating rates in relation to female body size and treatment, we constructed GLMs that used a logit link function and binomial error distribution, where the number of females that remated is the dependent variable and the total number of females assayed is the binomial denominator. We tested whether male mate preferences caused the tails of the distribution of female lifetime fecundity to regress towards the mean by performing a two-way ANOVA, with body size, remating treatment, and their interaction as predictor variables. A significant interaction term (that was associated with a smaller difference between the mean fecundity of large and small females when male preference was possible) would indicate that the tails of the fecundity distribution had regressed toward the mean. Each type of remating assay was repeated twice. The first, choice assay was comprised of ten adult competition vials (the unit of replication) for each body size treatment, whereas the second replicate was comprised of 38 adult competition vials in the large body size treatment and 37 in the small body size treatment. Both replicates of the no-choice assay were comprised of 15 adult competition vials for each body size treatment.

Acknowledgments

We thank the members of the Rice lab for helpful discussions. Comments from A. Agrawal, N. Gallo, and three anonymous reviewers also helped improve the manuscript.

Author Contributions

The author(s) have made the following declarations about their contributions: Conceived and designed the experiments: TAFL AP ADS WRR. Performed the experiments: TAFL AP ADS. Analyzed the data: TAFL WRR. Contributed reagents/materials/analysis tools: TAFL WRR. Wrote the paper: TAFL AP ADS WRR.

- 5. Amundsen T (2000) Why are female birds ornamented? Trends Ecol Evol 15: 149–155.
- Amundsen T, Forsgren E (2001) Male mate choice selects for female coloration in a fish. Proc Natl Acad Sci U S A 98: 13155–13160.
- Bonduriansky R (2001) The evolution of male mate choice in insects: a synthesis of ideas and evidence. Biol Rev Camb Philos Soc 76: 305–339.
- Jones IL, Hunter FM (1993) Mutual sexual selection in a monogamous seabird. Nature 362: 238–239.

- 9. Chenoweth SF, Petfield D, Doughty P, Blows MW (2007) Male choice generates stabilizing sexual selection on a female fecundity correlate. J Evol Biol 20: 1745 - 1750
- 10. Gwynne DT (1984) Sexual selection and sexual differences in Mormon crickets (Orthoptera: Tettigoniidae, Anabrus simplex). Evolution 38: 1011-1022.
- Gwynne DT, Bussiere LF, Ivy TM (2007) Female ornaments hinder escape from spider webs in a role-reversed swarming dance fly. Anim Behav 73: 1077-1082.
- 12. Gwynne DT, Simmons LW (1990) Experimental reversal of courtship roles in an insect. Nature 346: 172-174.
- 13. Parker GA (1979) Sexual selection and sexual conflict. In: Blum MS, Blum NA, eds. Sexual selection and reproductive competition in insects. New York (New York): Academic Press. pp 123-166.
- 14. Parker GA (1983) Mate quality and mating decisions. In: Bateson PPG, ed. Cambridge (Uited Kingdom): Cambridge University Press. pp 141-166.
- 15. Arnqvist G, Rowe L (2005) Sexual conflict. Princeton (New Jersey): Princeton University Press. 330 p.
- 16 Byrne PG, Roberts JD (1999) Simultaneous mating with multiple males reduces fertilization success in the myobatrachid frog Crinia georgiana. Proc R Soc Lond B Biol Sci 266: 717-721.
- 17. Davies NB, Halliday TR (1979) Competitive mate searching in male common toads. Bufo bufo. Anim Behav 27: 1253-1267.
- 18. Reale D, Bousses P, Chapuis JL (1996) Female-biased mortality induced by male sexual harassment in a feral sheep population. Can J Zool 74: 1812-1818.
- 19. Le Galliard JF, Fitze PS, Ferriere R, Clobert J (2005) Sex ratio bias, male aggression, and population collapse in lizards. Proc Natl Acad Sci U S A 102: 18231-18236.
- 20. Huxley JS (1912) A "disharmony" in the reproductive habits of the wild duck (Anas boschas,. L.). Biol Zent Bl 32: 621–623.
- 21. Fox EA (2002) Female tactics to reduce sexual harassment in the Sumatran orangutan (Pongo pygmaeus abelii). Behav Ecol Sociobiol 52: 93-101.
- 22. Arnqvist G (1989) Multiple mating in a water strider: mutual benefits or intersexual conflict? Anim Behav 38: 749–756.
- 23.Fowler K, Partridge L (1989) A cost of mating in female fruit flies. Nature 338: 760 - 761
- Partridge L, Fowler K (1990) Nonmating costs of exposure to males in female 94 Drosophila melanogaster. J Insect Physiol 36: 419-425.
- 25. McLachlan AJ (2006) You are looking mitey fine: parasites as direct indicators of fitness in the mating system of a host species. Ethol Ecol Evol 18: 233-239.
- 26. Fisher RA (1930) The genetical theory of natural selection. Oxford (United Kingdom): Clarendon Press. 272 p.
- Crow JF, Kimura M (1970) An introduction to population genetics theory. New 27. York (New York): Harper & Row. 591 p
- 28. Rice WR, Linder JE, Friberg U, Lew TA, Morrow EH, et al. (2005) Inter-locus antagonistic coevolution as an engine of speciation: assessment with hemiclonal analysis. Proc Natl Acad Sci U S A 102: 6527-6534.
- Rice WR, Stewart AD, Morrow EH, Linder JE, Orteiza N, et al. (2006) 29. Assessing sexual conflict in the Drosophila melanogaster laboratory model system. Philos Trans R Soc Lond B Biol Sci 361: 287-299.
- 30. Pitnick S, Garcia-Gonzalez F (2002) Harm to females increases with male body size in Drosophila melanogaster. Proc R Soc Lond B Biol Sci 269: 1821-1828.
- Fairbairn DJ, Reeve JP (1999) Change in sexual size dimorphism as a correlated response to selection on fecundity. Heredity 83: 697-706
- 32. Stewart AD, Morrow EH, Rice WR (2005) Assessing putative interlocus sexual conflict in Drosophila melanogaster using experimental evolution. Proc R Soc Lond B Biol Sci 272: 2029-2035.

- 33. Byrne PG, Rice WR (2006) Evidence for adaptive male mate choice in the fruit fly Drosophila melanogaster. Proc R Soc Lond B Biol Sci 273: 917-922.
- 34. Bateman AJ (1948) Intra-sexual selection in Drosophila. Heredity 2: 349-368. 35. Long TAF, Miller PM, Stewart AD, Rice WR (2009) Estimating the heritability of female lifetime fecundity in a locally adapted Drosophila melanogaster population. J Evol Biol 22: 637-643.
- Orteiza N, Linder JE, Rice WR (2005) Sexy sons from re-mating do not recoup 36. the direct costs of harmful male interactions in the Drosophila melanogaster laboratory model system. J Evol Biol 18: 1315-1323.
- 37. Stewart AD, Hannes AM, Mirzatuny A, Rice WR (2008) Sexual conflict is not counterbalanced by good genes in the laboratory Drosophila melanogaster model system. J Evol Biol 21: 1808-1813.
- 38. Morrow EH, Stewart AD, Rice WR (2008) Assessing the extent of genome-wide intralocus sexual conflict via experimentally enforced gender-limited selection. Evol Biol 21: 1946-1054.
- 39. Kamimura Y (2007) Twin intromittent organs of Drosophila for traumatic insemination. Biol Lett 3: 401-404.
- Wolfner MF (2009) Battle and ballet: molecular interactions between the sexes in 40. Drosphila, I Hered 100: 399-410.
- 41. Friberg U, Arnqvist G (2003) Fitness effects of female mate choice: preferred males are detrimental for Drosophila melanogaster females. J Evol Biol 16: 797-811
- 42. Chippindale AK, Gibson JR, Rice WR (2001) Negative genetic correlation for adult fitness between sexes reveals ontogenetic conflict in Drosophila. Proc Natl Acad Sci U S A 98: 1671-1675
- Lindsley DL, Grell EH (1968) Genetic variations of Drosophila melanogaster. 43. Washington, DC: Carnegie Institution. 472 p.
- 44. Gefen E, Marlon AJ, Gilbbs AG (2006) Selection for desiccation resistance in adult Drosophila melanogaster affects larval development and metabolite accumulation. J Exp Biol 209: 3293-3300.
- 45. Chenoweth SF, Doughty P, Kokko H (2006) Can non-directional male mating preferences facilitate honest female ornamentation? Ecol Lett 9: 179-184
- 46 Servedio MR, Lande R (2006) Population genetic models of male and mutual mate choice. Evolution 60: 674-685.
- 47 Arnqvist G, Nilsson T (2000) The evolution of polyandry: multiple mating and female fitness in insects. Anim Behav 60: 145-164.
- 48. Robertson FW, Reeve E (1952) Studies in quantitative inheritance. I. The effects of selection on wing and thorax length in Drosophila melanogaster. J Genet 50: 414-448.
- 49. Coyne JA, Beecham E (1987) Heritability of two morphological characters within and among natural populations of Drosophila melanogaster. Genetics 117: 727-735
- Ashburner M, Golic KG, Hawley RS (2005) Drosophila: a laboratory handbook, 2nd edition. Cold Spring Harbor (New York): Cold Spring Harbor Laboratory Press. 1409 p
- 51. Candolin U, Heuschele J (2008) Is sexual selection beneficial during adaptation to environmental change? Trends Ecol Evol 23: 446-452.
- 52. Lefranc A, Bundgaard J (2000) The influence of male and female body size on copulation duration and fecundity in Drosophila melanogaster. Hereditas 132: 243-247
- 53. Spieth HT (1974) Courtship behavior in Drosophila. Annu Rev Entomol 19: 385-405.
- Greenspan RJ, Ferveur JF (2000) Courtship in Drosophila. Annu Rev Genet 34: 205–232
- 55. Hall JC (1994) The mating of a fly. Science 264: 1702-1714.